Author’s response to reviews

Title: Longitudinal Study of Electrical, Functional and Structural Remodelling in an Equine Model of Atrial Fibrillation

Authors:

Eva Hesselkilde (evah@sund.ku.dk)
Helena Carstensen (hc@sund.ku.dk)
Mette Flethøj (flethoj@sund.ku.dk)
Merle Fenner (merle.fenner@sund.ku.dk)
Ditte Kruse (ditte.kruse@sund.ku.dk)
Stefan Sattler (sattler@sund.ku.dk)
Jacob Tfelt-Hansen (jacob.tfelt@regionh.dk)
Steen Pehrson (steen.pehrson@regionh.dk)
Thomas Hartig Braunstein (thobra@sund.ku.dk)
Jonas Carlson (jonas.carlson@med.lu.se)
Pyotr Platonov (pyotr.platonov@med.lu.se)
Thomas Jespersen (thojes@sund.ku.dk)
Rikke Buhl (rib@sund.ku.dk)

Version: 3 Date: 23 Sep 2019

Author’s response to reviews:

Dear Editor and reviewers,

First, we would like to apologies for the late re-submission. The editorial email was caught in the spam filter. We have adjusted the settings so it hopefully will not happen again!

To meet the request of Dr. Decloedt we have included the following sentence in L298: “however, as these analyses are underpowered similar conclusions cannot be made from the present study.”
In addition, the manuscript has been proofread again.
Furthermore, we have made the editorial changes and uploaded a clean manuscript.

Many thanks,

Rikke Buhl